Accelerator Status Report

Run Schedule: March 11, 2010 to August 13, 2010
July 28, 2010
Injector    Energy 61.88 MeV
· (7/27) GUN HV would not stay on, which was caused by the oil tank recently installed. The experts did work on it and improved the situation but it still trips.
· (7/20) Injector high current (160uA) setup was done for Qweak.

Accelerator   Linac Energy 550.00 MeV
· Over the last we did not deliver much beam due to issues with Hall B target. We used the time for Beam Studies and RF Q measurement.

· (7/27) Arc5, Arc6, Arc7, Arc8, Arc9, RSEP8A, RSEP9A, DOG5E, DOG7E and DOG9E are off because we only need pass 1 and pass 2.
· (7/26) BSY dropped to restricted access probably due to someone bump the exit door from the inside during

· (7/25) CHL tripped.

· (7/24) CHL tripped.

· (7/23) Leaking water hose to the master oscillator in the NL was repaired.

· None of the halls took beam for about nine days. The accelerator was used for beam studies and measurement of quality factors of the cavities.
· The CHL tripped with RF off Saturday owl shift. 

Hall A---Pass 2, APEX
· Installation of DVCS.

· (7/12) Hall A started the summer down.
Hall B ---Pass 2, g9b-Frost from March 11 to August 13, 2010
· (7/27) started to set up 2 pass beam to Hall B. It turned out that the beam size was 

      difficult to adjust at 2C21 location. We did ask for help. Several hours were spent and the

      desired beam size was achieved.

· The hall lost a lot of time due to issues with its target.

· Hall B finished the Micromegas Test.
· (7/20) Hall B started to take beam swing shift. They were down for about 12 days due to

       issues with its target. The hall is conducting Micromegas Test in FROST for CLAS12

      for two days. 
      Friday the hall will change back to the regular run with pass 2 beam (g9b-Frost)
Hall C--- Pass 1, QWeak from July 4 to August 13, 2010
· The hall plans to take beam Friday (7/29).

· Hall C will not be ready for beam until sometime next week due to the work on its targets.

